Author’s response to reviews

Title: A Case of Surgically-associated Anti GQ1b Antibody Syndrome Accompanied by Saccadic Ping Pong Gaze

Authors:

Jingzhe Han (hanjingzhe2017@sina.com)
Yanan Xie (xieyanan19860313@163.com)
Haiyan Yan (yanhaiyan1216@sina.com)
Yuecheng Cao (422521181@qq.com)
Hongmei Wang (30795969@qq.com)
Duanhua Cao (mengyu184@sohu.com)

Version: 1 Date: 07 Jul 2018

Author’s response to reviews:

Editor's comments:

1. Consent for Publication:

   a. Thank you for providing a statement on consent for publication. Unfortunately, we feel that "Written informed consent was obtained from the members of the patient or publication of this report" is not quite clear enough to proceed. Please can you confirm if written informed consent was obtained directly from the patient? If not, and it was obtained from a member of his family, please can you provide a reason for this?

Response: It has been revised in the manuscript.

   b. Please can you confirm that the consent covered the publication of the accompanying video?

Response: Confirmed.

2. Authors’ Contributions:

Please note that based on the current authors’ contributions section, several listed authors do not automatically qualify for authorship.
We ask that you ensure that author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Response: It has been revised in the manuscript.

3. Copy-edit:

I'm afraid the quality of the English used throughout your manuscript does not currently meet our requirements. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

Response: It has been revised by a native speaker.